

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Alzúa, María Laura; Cruces, Guillermo; Ripani, Laura

Working Paper Welfare programs and labor supply in developing countries: Experimental evidence from Latin America

IZA Discussion Papers, No. 6959

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Alzúa, María Laura; Cruces, Guillermo; Ripani, Laura (2012) : Welfare programs and labor supply in developing countries: Experimental evidence from Latin America, IZA Discussion Papers, No. 6959, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/67290

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 6959

Welfare Programs and Labor Supply in Developing Countries: Experimental Evidence from Latin America

María Laura Alzúa Guillermo Cruces Laura Ripani

October 2012

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

ΙΖΑ

Welfare Programs and Labor Supply in Developing Countries: Experimental Evidence from Latin America

María Laura Alzúa

CEDLAS-UNLP and CONICET

Guillermo Cruces

CEDLAS-UNLP, CONICET and IZA

Laura Ripani

Inter-American Development Bank

Discussion Paper No. 6959 October 2012

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 6959 October 2012

ABSTRACT

Welfare Programs and Labor Supply in Developing Countries: Experimental Evidence from Latin America^{*}

This study looks at the effect of welfare programs on work incentives and the adult labor supply in developing countries. The analysis builds on the experimental evaluations of three programs implemented in rural areas: Mexico's PROGRESA, Nicaragua's RPS and Honduras' PRAF. Comparable results for the three countries indicate that the effects that the programs have had on the labor supply of participating adults have been mostly negative but are nonetheless small and not statistically significant. However, the evidence does point to the presence of other effects on labor markets. In the case of PROGRESA, there is a small positive effect on the number of hours worked by female beneficiaries and a sizeable increase in wages among male beneficiaries and a resulting increase in household labor income. Moreover, PROGRESA seems to have reduced female labor-force participation in ineligible households. These results imply that large-scale interventions may have broader equilibrium effects.

JEL Classification: J08, J22, I38

Keywords: welfare programs, income support, labor supply, work incentives, conditional cash transfers, randomized control trials, developing countries

Corresponding author:

Guillermo Cruces CEDLAS-UNLP Centro de Estudios Distributivos, Laborales y Sociales Facultad de Ciencias Económicas Universidad Nacional de La Plata Calle 6 entre 47 y 48, 5to. piso, oficina 516 1900 La Plata Argentina E-mail: gcruces@cedlas.org

This study is based on a background paper entitled, "Labor supply responses to conditional cash transfer programs. Experimental and non-experimental evidence from Latin America", prepared for the Inter-American Development Bank (IDB). The authors wish to thank Santiago Levy for encouraging them to work on this study and Emanuel Skoufias for providing an early draft of his ongoing work. The authors also acknowledge financial support from the CEDLAS-IDRC research project on "Labor markets for inclusive growth in Latin America". The editor, Erdal Tekin, and an anonymous referee provided valuable feedback. Comments by Felipe Barrera, Sami Berlinski, César Bouillon, Sebastián Galiani, Laura Guardia, Pablo Ibarrarán, Miguel Jaramillo, Julia Johannsen, Santiago Levy, Florencia López Boo, Craig McIntosh, Claudia Piras, Patrick Puhani, Graciana Rucci, Norbert Schady, Guilherme Sedlacek, Ana Santiago and Yuri Soares are much appreciated. We also gratefully acknowledge the comments received from participants at the 13th Annual Meeting of the Latin American and Caribbean Economic Association (LACEA) in Rio de Janeiro in 2008, at the AfrEA-NONIE-3ie Impact Evaluation Conference in Cairo, held in April 2009, and at annual conference of the North East Universities Development Consortium (NEUDC) held in 2009. Andrés Ham and Nicolás Epele provided outstanding research assistance. The usual disclaimer applies. The opinions expressed in this report are those of the authors and do not necessarily represent those of the institutions to which they belong.

1 Introduction

This study explores the effect of welfare programs on work incentives and the adult labor supply in developing countries. The analysis builds on the experimental evaluations of three programs implemented in rural areas in Latin America: Mexico's PROGRESA, Nicaragua's *Red de Protección Social* ("Social Protection Network") (RPS) and Honduras' *Programa de Asignación Familiar* ("Family Allowance Program") (PRAF). The study takes advantages of the random assignment of localities to program deployment and control groups and presents comparable estimates of impacts on the adult labor supply and remuneration levels. These estimates are based on homogeneous datasets and were arrived at through the use of a common estimation methodology.

The impact of welfare and income-support programs on labor supply has been widely studied in developed countries (Moffitt, 2002; Meghir and Phillips, 2008; Moffitt and Scholz, 2009). This literature has pointed up the existence of work disincentives among recipient households, and these and other considerations have prompted recent reforms that have incorporated sophisticated measures to mitigate these negative effects (Moffitt, 2003a; Blundell and Hoynes, 2004; Dickens et al., 2004; Michalopoulus et al., 2005). The programs under study here are conditional cash transfer (CCT) programs, which combine monetary benefits with incentives for curbing child labor and fostering the accumulation of human capital. Benefit receipt is subject to a series of verifiable conditions, such as school attendance, vaccination and regular medical checkups, among others. The results of a number of evaluations in Latin America indicate that cash transfers, especially when combined with conditionalities, have proved successful in increasing welfare and human capital accumulation in recipient households and in reducing child labor (see the reviews by Rawlings and Rubio, 2003 and 2005; and Fiszbein and Schady, 2009).

Unlike their recent counterparts in the United States and Europe, however, these programs do not incorporate measures to guard against potential negative impacts on the adult labor supply. Moreover, there is very little consistent, systematic evidence regarding this aspect, despite the existence of a wealth of empirical analyses concerning their intended outcomes. This study attempts to establish whether these cash transfers have any incentive effects on the labor supply of adults in recipient households, on non-eligible individuals and on the broader labor market equilibrium.

The main contribution made by this study is the systematic, clear-cut evidence that it provides concerning the labor-supply effects of welfare programs in developing countries.

Despite the crucial role played by such programs in the income-generation process among poor segments of the population, there is limited evidence concerning labor-supply decisions in this context. Existing studies point to the presence of complex interactions among public policy, work incentives and labor allocation within households (see, for instance, Ardington et al., 2009). Moreover, the systematic evidence presented below is derived from experimental evaluation designs,¹ which have clear advantages over the policy and natural experiments underlying most previous studies of welfare programs and labor outcomes (Angrist and Krueger, 1999; Blundell and MaCurdy, 1999; Imbens et al., 2001; Eissa et al., 2008). These evaluation strategies have also overcome some of the shortcomings of previous randomized experiments, such as those of the negative income tax of the 1960s and 1970s in the United States (Ashenfelter and Plant, 1990; Moffitt, 2003b).

Comparable results for the three countries indicate that the effects of these programs on the labor supply of participating adults are, while primarily negative, small and nonsignificant. Even though they provided considerable transfers, the programs did not reduce the labor supply substantially in the short term. However, the evidence also reveals the presence of other effects on labor markets. In the case of PROGRESA, there was a small positive effect on the number of hours worked by female beneficiaries, a sizeable increase in wages, especially among male beneficiaries, and a resulting increase in household labor income after the program had been in operation for two years. These impacts can be attributed to changes in the labor supply of adults in eligible households, and to the increased amount of time available to women as a result of higher school enrollment rates among children. Moreover, PROGRESA seems to have reduced the female labor-force participation rate in ineligible households in the localities randomly assigned to the program. This mechanism is related to recent findings on the indirect impact of CCTs on ineligible households (Angelucci and De Giorgi, 2009) and implies that large-scale interventions can have broader equilibrium effects. This additional layer of complexity should be considered in the design and evaluation of future interventions. These equilibrium effects also have important consequences for the interpretation of results from randomized controlled trials (see Moffitt, 2003b; Duflo et al., 2008; Heckman, 2008; and the debate between Deaton, 2009, and Imbens, 2010).

¹ Behrman and Todd (1999), Skoufias et al. (1999), Skoufias and Parker (2001) and Gertler (2004), among others, describe the original experimental evaluation strategy of Mexico's PROGRESA on which the evaluations of RPS and PRAF were based.

The study is organized as follows. Section 2 discusses the theoretical underpinnings of the potential impact of cash transfers on labor supply and presents a review of the empirical evidence for countries in Latin America. Section 3 briefly reviews the programs and their evaluation strategies, and then goes on to describe the estimation and inference procedures. Section 4 presents the empirical results on labor market outcomes for adults in the three programs. Section 5 provides an overview of the results and some conclusions.

2 Labor markets and conditional cash transfer programs

2.1 Potential impacts of CCTs on labor markets

CCT programs combine short-term poverty alleviation (through cash transfers) with long-term outcomes that are achieved through the use of incentives for human capital accumulation (school attendance, health check-ups, improved nutrition and the reduction of child labor).² With the exception of minor training components in some programs, the overall design of the CCTs in Latin America is not directly related to the employment of adults in beneficiary households. There are no restrictions on work, and unlike previous workfare-like initiatives in developing countries, CCTs do not use low-wage jobs as targeting mechanisms (Besley and Coate, 1992; Kanbur et al., 1994). Most importantly, earned labor income does not reduce benefit levels. In this sense, CCTs constitute a simpler policy instrument than welfare programs in developed countries: as a pure subsidy (as far as adults are concerned – although some of the conditionalities might imply some costs in terms of time), CCTs do not induce steep replacement rates as traditional welfare programs do, nor do they entail the complexity of welfare-to-work initiatives such as the Earned Income Tax Credit in the US (Eissa and Liebman, 1996) or the UK's Working Families' Tax Credit (Meghir and Phillips, 2008).³

The lack of work requirements does not mean, however, that the programs are neutral in terms of adult labor supply and work incentives. The income-support component and the conditionalities relating to children's health and education might still have affected these outcomes. Economic theory suggests several ways in which CCTs can affect work decisions

² A more detailed description of the three programs is provided in Section 3.1 and the appendix.

³ Moreover, program overlap is less of a problem for program evaluation (Moffitt, 2002) in the cases under study: PROGRESA consolidated several different programs in Mexico, while PRAF and RPS represented some of the first attempts made to provide widespread income support in Honduras and Nicaragua.

within recipient households. In a standard static model of choice between consumption and leisure, the components of CCTs may play a role through at least four channels.

Firstly, the cash transfer component of the program generates an increase in unearned non-labor income. As such, it induces a pure income effect, which loosens the budget constraint of the recipient households. The rise in unearned income could reduce the number of hours of work if leisure is a normal good for beneficiaries, but the presence of fixed hour or money costs, such as commuting or childcare (Cogan, 1981; Bhattarai and Whalley, 2003; Kluve and Tamm, 2012), may induce an increase in labor supply as a result of the lump-sum transfer (Ralitza and Wolff, 2011).

Conditionalities constitute the second channel through which CCTs may induce behavioral responses in the adult labor supply. The requirements related to children's human capital accumulation may have an impact on a household's allocation of time: the positive impact of CCTs on children's school attendance could free up time previously spent on childcare (Blau and Tekin, 2007; Baker et al., 2008; Mörk et al., 2011), thereby further reducing the cost of work. There is some evidence of reduced participation of women in domestic work in the case of PROGRESA (Parker and Skoufias, 2000).

The third channel is related to the potential decrease in household income associated with the reduction in child labor (Basu and Hoang Van, 1998). This effect diminishes the net impact of cash transfers in households where children are induced to reduce their participation in work activities and could therefore mitigate the transfer's potential disincentive in terms of the adult labor supply.⁴

Finally, the fourth channel operates through different types of spillovers. On the one hand, there may be indirect effects: Angelucci and De Giorgi (2009), for instance, find that PROGRESA has had an impact on the consumption of ineligible households in program communities, and Bobonis and Finan (2009) report substantial spillovers in terms of secondary-school enrollment decisions for the same program. On the other hand, changes in the labor supply schedule of beneficiaries may affect aggregate wage levels and thus remunerations for both recipients and non-recipients. In the presence of such effects, the identification strategy based on the random allocation of the program would be partially compromised owing to a violation of the stable unit treatment value assumption (Angrist et

⁴ Yang's (2008) results for remittances in the Philippines do not point up any significant impacts of windfall income on the adult labor supply. However, the findings of Ardington et al. (2009) concerning migration from South Africa indicate that transfers may influence even more complex within-household interactions, thereby inducing unexpected labor-supply responses.

al., 1996). In terms of the labor supply, equilibrium effects reduce the scope for the interpretation of reduced form estimates as simple labor supply elasticities with respect to unearned income.

The combination of these four channels implies that the overall effect of CCTs on labor market outcomes for adults is ambiguous from a theoretical point of view. The presence of any impact, and its direction, is ultimately an empirical question.

2.2 The impact of CCTs on labor markets: previous findings for Latin America

Most of the literature on the impact of CCTs focuses on the programs' intended outcomes. While results vary from country to country, program evaluations reveal, to some degree, a positive effect on years of schooling, reductions in child labor and improvements in some key health indicators (Rawlings and Rubio 2003 and 2005; Bouillon and Tejerina, 2006; Fiszbein and Schady, 2009), as well as other related unintended effects on, for instance, fertility (Todd et al., 2011).

Effects on the adult labor supply have been partially analyzed for PROGRESA and RPS. The significant reduction in child labor found in the case of PROGRESA (Skoufias and Parker, 2001) contrasts with the absence of an impact on labor market outcomes for adults in beneficiary households, according to results from Parker and Skoufias (2000) and Skoufias and Di Maro (2008). Both studies, based on probit estimations, find no significant program effects on adult labor-force participation within eligible households in program localities. Also in the context of PROGRESA, Angelucci and De Giorgi (2009) find that household equivalent labor earnings for adults are not affected by the program. This study provides further results from estimations that include fixed effects at the household or individual levels. While these controls may not be strictly necessary in the context of a randomized controlled trial, they allow for better controls for baseline differences in employment (see Table A2 and Behrman and Hoddinott, 2005) and may also induce a gain in precision (Duflo et al., 2008). The results include these additional controls and confirm the robustness of previous findings.

The impact of Nicaragua's RPS program on the adult labor supply is analyzed in detail by Maluccio (2007). While the studies on PROGRESA referred to in the previous paragraph concentrated on individual labor-force participation and household earnings, Maluccio (2007) studies the effect of RPS on total hours of work at the household level. The

results, obtained by means of a random effects model, indicate that the program has had a small but significant negative effect on total household hours of work, with most of the negative impact relating to the amount of time spent in agricultural activities. These effects and their causes are discussed in detail below, with the evidence being presented here pointing to a household composition effect rather than a direct effect on hours worked.

There are fewer papers that draw upon evaluation data for Honduras' PRAF program. Galiani and McEwan (2012) developed an original evaluation strategy based on census data instead, which was collected shortly after the program was implemented. They report no significant effects for PRAF on the labor supply of adult women and only a small decrease (1 percent) for adult males, although this estimate is not robust to alternative specifications.

The analysis presented in this study provides comparable results for the three programs. They are based on a common procedure for processing the original datasets, which leads to homogeneous definitions for dependent and independent variables. Moreover, the estimates for the three programs are derived from the same methodology and allow for the same type of controls for randomization imbalances and other issues by including individual and household fixed effects. Finally, while evaluations of PROGRESA and PRAF have concentrated on individual participation and those of RPS on household hours, the results detailed below allow for further disaggregation in order to look at participation, hours of work, sector allocation, household labor earnings and wages (when possible) for all programs.

3 Experimental evaluation strategies and estimation methodology

3.1 The programs and their evaluations

The data used in this analysis are drawn from ad hoc longitudinal surveys carried out in order to evaluate each specific intervention. The three programs share a common evaluation methodology, with baseline and follow-up data collection being conducted in localities that were randomly assigned to program deployment and in those that were selected into the control group. The three data sources were harmonized on the basis of a common set of criteria in order to achieve maximum comparability using the methodology described in CEDLAS (2012). The three interventions targeted rural areas in poor regions of the respective countries. The following paragraphs briefly describe the three programs' evaluation strategies,⁵ based on PROGRESA's experimental design, which randomized program deployment at the locality level.

In 1997, Mexico began implementing the first phase of PROGRESA. It was geographically targeted by locality. From an initial group of the 506 localities that were selected for the first round, 320 were randomly selected to participate in the program, which was not deployed in the remaining 186 localities. Households in the latter localities were still subject to the data collection process and thus constituted the control group for the program's evaluation. The intervention also included a targeting rule based on a proxy means test: only qualifying households in treatment localities were eligible to participate.

The data employed in this study are drawn from the PROGRESA Evaluation Survey (ENCEL). The estimates discussed below are based on the initial baseline survey and on three follow-up rounds⁶ conducted at six-month intervals following program implementation. The surveys collected socio-demographic and labor-market information for all households and individuals in both treatment and control communities.

Honduras' PRAF was implemented in a set of 50 randomly selected municipalities of a total of 70, with the 20 remaining municipalities forming the control group. The data used in this study correspond to a baseline survey carried out in the second half of 2000 and a follow-up survey in 2002. In contrast with PROGRESA, where all households in treatment and control localities were interviewed, the PRAF surveys covered only a representative sample of households. The corresponding sampling weights are used in the empirical work outlined below.

For the case of Nicaragua's RPS, half of the countries' poorest 42 localities were randomly assigned to the treatment group. The data used in this study are drawn from the initial baseline survey carried out in the third quarter of 2000 and the first and second follow-up surveys conducted in October 2001 and October 2002, respectively. As with the PRAF evaluation data, the survey consists of a representative sample of the population in treatment and control localities, and sampling weights are used for the estimations.

⁵ The structure of each program is detailed in the appendix to this study. Further references may also be found in Todd (2004) for PROGRESA, Glewwe and Olinto (2004) for PRAF, and Maluccio and Flores (2005) for RPS.

⁶ Baseline data were gathered between November 1997 and March 1998. The first, second and third follow-ups correspond to November 1998, March 1999 and November 1999, respectively.

Finally, although the programs have a number of characteristics in common, it should be noted that there were significant differences in the average size of the cash transfers provided by each of the initiatives. Imputing transfer values from each program's eligibility rules to the evaluation samples used in this analysis, the transfers represented about 4 percent of total household consumption for PRAF, 20 percent for RPS and 40 percent for PROGRESA.⁷ The potential effect of these differences is discussed in the section that covers the empirical results.

3.2 Estimation and inference with random assignment by locality

In view of the random assignment of localities in the context of the three programs under study and the availability of repeated observations, a differences-in-differences (DD) estimation technique is the most suitable one of exploiting the evaluation design and identifying the causal effects of the programs. A standard DD model with controls takes the form of:

$$Y_{ist} = A_s + B_t + cX_{ist} + \beta I_{st} + \varepsilon_{ist}$$
(1)

where Y_{ist} denotes the outcome variable of interest for individual (or household) *i* in group (or village) *s* at time *t*, I_{st} is an indicator variable representing treatment status for group *s* in time *t* (or alternatively, an interaction between a treatment group indicator and time effects), A_s and B_t are group and time effects, respectively, X_{ist} is a matrix of individual characteristics and ε_{ist} is an error term. The estimate of the program impact is the coefficient β . Without the X_{ist} controls and with two time periods, the estimate of β by ordinary least squares (OLS) is simply the difference in changes in mean outcomes between the treatment and control groups between the two time periods. The more general case, with more than two time periods, adds a full set of time controls and interactions to account for differential evolutions over time.

The canonical DD model of equation 1 without including individual controls X_{ist} provides estimates of β that amount to differences in the outcomes at the locality level. The evaluation of PROGRESA, PRAF and RPS, however, collected repeated household and individual observations, which means that a much richer set of information is available and

⁷ These estimates are roughly in line with others given in the literature: Maluccio (2004) reports 4 percent for PRAF, 18 percent for RPS and 20 percent for PROGRESA, although, for the latter, Gertler (2004) computes the average transfer as one third of total household income.

can be exploited (Wooldridge, 2001 and 2007). Specifically, the inclusion of individual (or household) fixed effects in the estimation of equation 1 permits the identification of program effects while controlling for some of the pre-treatment differences between localities (see Table 1, discussed below, and the Appendix). Moreover, this allows for a potential gain in precision (Duflo et al., 2008). While these individual fixed effects were not accounted for in the studies of labor supply reviewed in the previous section, they are routinely included in evaluations of CCT impacts on other outcomes (for instance, in Gertler's 2004 evaluation of PROGRESA's effect on health, among many others). The results discussed below provides two sets of estimates for each outcome based on equation 1: one with a full set of individual controls X_{ist} , and one with a full set of individual fixed effects but no time invariant X_{ist} variables.

With respect to the estimation methodology, the empirical results presented below are based on linear models – either OLS or fixed effect (FE) estimations of equation 1 – for binary dependent variables such as labor-force participation and for continuous variables such as hours of work, wages and income. As pointed out by Angrist and Pischke (2008), linear probability model estimates do not differ substantially from those of probit or logit regressions. Moreover, coefficients for the indicator and interaction variables in equation 1 have a straightforward causal interpretation for linear estimates.

All the results presented below give estimates of β in equation 1 over the full treatment and control samples, which correspond to intention to treat (ITT) coefficients. In the case of PROGRESA, the dataset contains a multidimensional targeting score which was used as a proxy means test for participation within program localities, thus making it possible for the eligibility status of each household to be known. For this reason, PROGRESA's results are also computed as differences between eligible households in treatment and control localities (average treatment effect – ATE),⁸ and differences between ineligible households in the two sets of localities. The latter estimates correspond to Angelucci's and De Giorgi's (2009) indirect treatment effects (ITE).⁹ To account for potential heterogeneous effects of the programs, the estimations are also computed by conditioning on the gender of the individual

⁸ Since take-up was very high among eligible households, average treatment effects and average treatment effects on the treated are roughly equivalent (Angelucci and De Giorgi, 2009). For simplicity, the ATE terminology is adopted in the description of the results.

⁹ Angelucci and De Giorgi (2009) exclude from their analysis a subset of those deemed ineligible in the initial phase of the program because of later changes in the eligibility rules. The analysis here follows Duflo et al. (2008) in exploiting only the primary assignment process regardless of changes in the program rules after the initial stage.

or the household head, as an alternative to the inclusion of multiple interactions (Djebbari and Smith, 2008).

Finally, the standard errors in the estimations need to account for the structure of the programs' evaluation and implementation processes. In the context of the three CCTs under study, random assignment did not apply directly over beneficiary households or individuals. The allocation was instead done at the geographical level. In terms of the equation above, randomization occurs at the group (village) level (*s*) instead of the individual or household level (*i*). Since eligibility for the program is defined at the group level, the standard errors of the DD estimates should account for the likely intra-cluster correlation to avoid a potential bias. Donald and Lang (2007) attribute this bias to the fact that many of the outcomes analyzed in the literature are serially correlated, which is not usually controlled for in DD estimations (see the discussion in Bertrand et al. (2004 – BDM henceforth)). This issue may be particularly important in the case of the labor-market outcomes covered in this study. A failure to account for this correlation across the randomization groups makes the usual OLS standard errors inconsistent and leads to erroneous inferences of the program's causal effects.

BDM propose two methods to correct the standard errors of estimates in equation $1:^{10}$ (i) taking into account the serial correlation of the outcome variable in each group *s* (this is known as cluster-robust variance estimation (CRVE) and is implemented by clustering observations by the assignment groups (e.g., localities)); and (ii) estimating standard errors using block bootstrap with replacement. The first method was used to arrive at the empirical results presented below; the standard errors are virtually equal to those obtained from block bootstrapped standard errors.¹¹ These corrections to the covariance matrix yield unbiased estimates of household- or individual-level outcomes in geographic targeting settings, thus accounting for potential serial correlation across groups.

¹⁰ BDM also propose a third correction that involves aggregating the data into group-year cells and estimating this model. However, only results from individual-level data are reported below.

¹¹ The working-paper version of this document (Alzua et al., 2010) presents the two sets of standard errors, with estimates following Cameron's, Gelbach's and Miller's (2008) suggestion of reporting bootstrapped CRVE-corrected standard errors.

4 CCTs and labor-market outcomes for adults

4.1 Descriptive statistics and random assignment processes

This section discusses the empirical evidence regarding the effect of CCT programs on labor outcomes for adults using experimental evaluation data from the three interventions detailed above: PRAF (Honduras), PROGRESA (Mexico) and RPS (Nicaragua).

Table 1 presents a series of descriptive statistics compiled at the time of the baseline survey for both treatment and control localities for these three programs. These statistics make it possible to verify the nature of the balance between the treatment and control groups in terms of observables. As expected in a rural setting in developing countries, household size in all three programs is fairly large, with an average of more than six individuals per household. About 70 to 80 percent of these households include two spouses. The calculations in the tables show that treatment and control households are not significantly different in terms of their demographic composition, with only a few small significant differences for some variables.

[TABLE 1 ABOUT HERE]

Table 1 also presents average educational levels for the treatment and control localities in each program. Since the programs are targeted at poor areas in each country, the distribution of educational outcomes is concentrated in lower levels of attainment, with about 5 years of education for PROGRESA, 2.2-2.3 for RPS and 3.4-2.9 for PRAF. The differences in educational achievement and enrollment rates between the treatment and control groups are small, except in the case of PRAF.

Finally, with respect to labor-market outcomes, the program datasets allow for nothing more than a simple definition of participation: individuals report if they work or if they do not. Employment ranges from 51.2-53.1 (PROGRESA) to 56.2-57.7 (RPS) and to 66.3-64.5 (PRAF). It is substantially higher for men than for women in the samples for the three countries (a difference of about 55 percentage points in PRAF and of about 70 percentage points in PROGRESA and in RPS). Employment is also higher in households with children and in single-headed households.

The unconditional means of socioeconomic and demographic statistics indicate some pre-program differences between treatment and control groups at the individual and household levels.¹² These results are generally in line with preexisting reports on these programs, which also found some significant differences between treatment and control localities (see Behrman and Todd, 1999, for PROGRESA; Glewwe and Olinto, 2004, for PRAF; and Maluccio and Flores, 2005, for RPS). Given the nature of the random assignment process in the three programs, implemented at the locality level, these differences probably arise because of the small number of effectively randomized units (localities). These differences reflect the composition of the resulting samples rather than their selection into treatment. In any case, the estimations discussed below control for individual characteristics or, alternatively, for individual fixed effects to account for the ex post differences in the treatment and control samples.

4.2 The effect of CCTs on labor-market outcomes for adults

The analysis of labor-market outcomes for adults in the three programs is restricted to a common sample selection criterion which includes individuals between 15 and 80 years old. Estimates of household-level outcomes are restricted to household heads in the same age range.

The original evaluations focused primarily on each program's intended outcomes, such as children's health and education. The evaluation surveys have a much smaller set of labor-market indicators than larger periodic surveys use. In the three data sources employed in this study, the adult population can be divided into two alternate categories of labor-market status: those who work outside the home and those who do not.¹³ The discussion refers to work, employment and labor supply interchangeably.

There are other labor-market outcomes of interest, besides employment status, that can be explored using these evaluation datasets: the number of hours worked in all occupations in a week (for those with positive hours); an indicator for employment in agricultural activities (for those employed); and the total hours worked in the household by members from 15 to 80 years of age (this variable is computed and estimated at the household level for households with positive hours).

¹² This is also apparent in a conditional framework, as discussed in the appendix in respect of the analysis of the random assignment process, which indicates that the resulting treatment and control localities have significant differences in some dimensions for the three programs.

¹³ It is thus not possible to distinguish between inactivity and unemployment. This distinction is feasible for the RPS data, but in the interests of comparability, the results detailed below are reported for the same variable for the three programs.

As stated in the previous section, the results correspond to two alternative specifications for each outcome of interest. On the one hand, the tables report OLS estimates β in equation 1 with a series of controls:

- Controls for individual characteristics: gender (if applicable), household size, an indicator for two-parent households, number of children, age of the individual, age squared, and educational indicators (complete primary through complete university).
- Controls for household characteristics: the gender of the household head (if applicable), household size, an indicator for two-parent households, number of children of the head of household, a dummy variable indicating if at least one child in the household attends school, and indicators for the household head's educational level.

The fixed-effect (FE) estimations, on the other hand, do not include any individual controls, since most of those listed above are time invariant or have low variability. All estimations include time effects, treatment indicators, interactions between the two, and locality controls, with standard errors clustered at the locality level. Finally, the results present estimates for the intention-to-treat (ITT) for the three programs and for males and females separately. For the specific case of PROGRESA, the availability of eligibility status data means that average treatment effects (ATE) and indirect treatment effects (ITE) can also be computed. The tables only report the relevant coefficient for the treatment effects (the coefficient β).

The estimates for PRAF correspond to the simple two-period case (baseline in second half of 2000, follow-up in May-August 2002), while estimates for RPS and PROGRESA include multiple consecutive follow-up surveys.¹⁴ The RPS baseline was established in August-September 2000, with a first follow-up in October 2001 and a second one in October 2002. For PROGRESA, the baseline corresponds to September 1997-March 1998, while the follow-up data were collected in November 1998, March 1999 and November 1999.

The evidence concerning the main theoretical question – the impact of each program on employment – is presented in Tables 2 and 3, which show the estimated coefficient of the treatment period/treatment status interaction in equation (1). Table 2 presents the results for Honduras' PRAF and Nicaragua's RPS programs. None of the estimates of the programs' effects on employment are statistically different from zero at standard significance levels. The estimates range from -0.5 to -1.8 percentage points for PRAF and from -0.2 to -2.3

¹⁴ The tables report the effect by round of the evaluation survey and correspond to the difference between the round and the baseline (pre-program) levels. These effects are estimated jointly by multiple time and treatment interactions, not as separate regressions by follow-up period.

percentage points for RPS (with positive effects of 0.6-0.9 points for males in the first followup survey). In all cases, the effects are higher in terms of absolute value for females than for males.

[TABLE 2 ABOUT HERE]

Table 3 presents the results for PROGRESA. The coefficients on employment, estimated jointly for males and females, are also negative and are in a similar range to those reported in Table 2 (from about -0.3 to -2.6 percentage points). Despite some statistically significant coefficients, there does not appear to be a consistent pattern of significant results for all three follow-up periods or for both OLS and FE estimations. The overall and average treatment effects are compatible with a setting in which income effects are either small or counterbalanced by other forces, as discussed in Section 2.

[TABLE 3 ABOUT HERE]

However, the ITE estimates (effects on individuals in ineligible households) exhibit a higher degree of significance, which seems to be driven mostly by a large fall in employment among ineligible females in the third follow-up survey (about 3.5-4.9 percentage points, for FE and OLS estimations, respectively). While this result may be a statistical artifact, additional results regarding type of employment, hours and household labor income indicate that there may be composition effects within households and between eligible and ineligible individuals. These overall effects are discussed in detail below following the presentation of the rest of the empirical evidence.

Besides their effect on overall employment, the programs may also affect occupational choice. For instance, Skoufias et al. (2008) find that the *Programa de Apoyo Alimentario* ("Food Support Program) (PAL) program in Mexico induced workers to move away from agricultural work, which supports the idea that this kind of work acts as food insurance. Tables 4 and 5 present the results for regressions for employed individuals, in which the dependent variable is an indicator equal to one if they work in agricultural occupations and is zero otherwise.

[TABLE 4 ABOUT HERE]

The coefficients reported in Table 4 indicate that neither PRAF nor RPS induced a substantial shift in labor allocation to agricultural or other sectors at the aggregate level. The coefficients for the overall population are negative for PRAF, and their sign for RPS depends on the estimation method that is used, but they are not statistically significant for either of the two programs. Table 5, however, indicates a positive, significant and large effect on agricultural employment in Mexico for ineligible males in the second and third follow-up

rounds (of 6.1 and 4.5 percentage points, respectively), although these effects are statistically different from zero only for the FE estimates. In contrast, the average treatment effect for males (the effect on those eligible) is substantially closer to zero (ranging from -0.8 to 1.5 percentage points) and not significant at standard levels for any of the three treatment periods. This reinforces the existing evidence of the presence of composition effects by household and by eligibility status.

[TABLE 5 ABOUT HERE]

A more detailed picture of labor-market effects emerges from the analysis of Tables 6 and 7, which present regressions in which the dependent variable is the number of hours worked for individuals with strictly positive reported hours. The estimates for PRAF are consistently positive and small (from about 0.5 to about 1.9 hours per week), while those for RPS are consistently negative (from about -1.5 to about -5.7) and are higher in terms of absolute values for women (-3 to -5.7, depending on the follow-up and estimation method). However, none of the estimates for PRAF and RPS in Table 6 are significantly different from zero.

[TABLE 6 ABOUT HERE]

The estimates for PROGRESA (Table 7) are available for the first and third follow-up (information on hours worked was collected only for these surveys). The effects for all adults are substantially smaller than they are for PRAF and RPS, and they are not statistically significant for the full sample or for eligible and ineligible males.¹⁵ However, there is a small but consistently significant positive average treatment effect of about 0.4 additional hours worked per week by female beneficiaries for the two available follow-up periods (ATE with OLS and FE estimates) and a smaller but still significant ITT estimate of 0.18-0.36 hours in the third follow-up. These results are for working individuals and indicate a small adjustment in the intensive margin of labor supply for women, which is compatible with the idea that beneficiaries have more time available than before because of the increase in children's school enrollment documented for PROGRESA.

[TABLE 7 ABOUT HERE]

While the former effect refers to the impact of the programs on individual hours of work, Tables 8 and 9 present the effects on total hours of work by adults in the household, per adult (these are household, not individual, estimates). The results for PRAF in Table 8 are

¹⁵ Angelucci and De Giorgi (2009) similarly fail to find significant effects on hours worked for non-eligible individuals in PROGRESA.

similar to those shown in Table 6, with small and positive coefficients (in female-headed households, the coefficients are larger for OLS estimations), but the overall effects on number of hours worked are not significant. The results for RPS are also similar to those given in Table 6: there are larger negative effects in terms of the number of hours worked per adult, which was higher by the time of the first follow-up survey (a year after the baseline survey), but these estimates are not significantly different from zero. The results for RPS, however, differ from those reported by Maluccio (2007), who finds a significant fall in the hours worked by adults. The difference between Maluccio's (2007) estimates and those presented here is that the dependent variable in the regressions reported in Table 8 is the total number of hours worked by adults in the household per adult, while Maluccio (2007) uses total overall hours for the household. Replicating Maluccio's (2007) estimates indicates that there is indeed a negative and significant effect on total hours at the household level, but this is driven by a household composition effect: the number of adults in households fell significantly in female-headed households in RPS.¹⁶ The results for PROGRESA in Table 9 indicate small and not statistically significant results for this household aggregate, even in female-headed households.

[TABLE 8 ABOUT HERE]

The overall results indicate that the programs did not introduce substantial disincentives to work, with no significant effects on the intensive or the extensive margin of labor supply for individuals and households in treatment localities. The small but significant increase in hours of work for female beneficiaries in PROGRESA is compatible with the presence of other factors that counterbalance the income effects, as discussed in Section 2.

[TABLE 9 ABOUT HERE]

However, the empirical results indicate that PROGRESA led to a substantial reduction in employment levels for ineligible women and a shift among ineligible men toward work in agricultural activities. These results are compatible with Angelucci's and De Giorgi's (2009) evidence on PROGRESA's effects on ineligible individuals. The following section deals with the effect of PROGRESA on wages and labor income and provides a fuller picture of the program's effect on labor-market outcomes for adults.

¹⁶ These additional results for RPS are presented in Table A4 in the Appendix.

4.3 The effect of PROGRESA on wages and household labor income

The discussion in Section 2 highlighted the possibility that cash transfer programs such as PROGRESA can have equilibrium effects by, for instance, shifting the aggregate labor supply curve by withdrawing children from the labor market, by freeing up adults' time, or by changing the latter's willingness to work through an income effect. It may also change relative remuneration levels, for instance by changing the sector allocation balance between beneficiaries and non-beneficiaries, as observed for PROGRESA in Table 5. The PROGRESA dataset provides a basis for an analysis of the program's effects on individual hourly wages and on household labor income per adult (this information was not collected by the PRAF and RPS evaluation surveys).

The results of the regressions presented in Table 10 indicate that PROGRESA had a sizeable effect on hourly wages in the treatment localities, although this effect seems to be driven mainly by eligible males (none of the coefficients for females are statistically significant, and none of the indirect treatment effect estimates are either). The ITT estimates indicate an increase in hourly wages of about 5.7 percent by the time of the third follow-up survey, with a higher average treatment effect coefficient of about 6.9 percent (both coefficients are significant only for the FE estimates). When restricting the sample to males, the ITT and ATE FE estimates indicate an effect of about 7.5 and 9.8 percent, respectively.

[TABLE 10 ABOUT HERE]

Finally, these higher hourly wages are partially reflected in higher levels of household labor income per adult. This effect is reported in Table 11, which indicates an increase of about 3.9 to 4.6 percent (for FE estimations, ITT and ATE, respectively), which is concentrated in the third round of the follow-up (two years after the baseline survey) and among male-headed households. These effects, however, are statistically significant only at the 10 percent level and then only for fixed-effects ITT and ATE estimates. This evidence on labor income for adults is not incompatible with Angelucci's and De Giorgi's (2009) results for monthly adult equivalent labor earnings in PROGRESA.¹⁷ These effects on individual

¹⁷ Angelucci and De Giorgi (2009, see Table 5) report that PROGRESA's average and indirect treatment effects on monthly adult equivalent labor earnings were not significant, based on results obtained by unconditional difference in differences estimation. However, they state in the notes to this table that they found a positive and significant (at the 10 percent level) average treatment effect for the third-round estimate when they included conditioning variables in their regressions. This finding is compatible with the result reported in Table 11 in this study, which includes individual controls (for OLS regressions) and individual fixed effects (for FE regressions).

wages and household earnings are discussed in the context of the overall results for PROGRESA in the following section.

[TABLE 11 ABOUT HERE]

5 Discussion and conclusions

This study of the effect of welfare programs on work incentives and the adult labor supply in developing countries is based on estimates derived from the experimental evaluations of three programs implemented in rural areas: PROGRESA in Mexico, RPS in Nicaragua and PRAF in Honduras.

The empirical results indicate that none of the three CCT programs has had any major impact on labor-market outcomes for adults, with no discernible effects on any of the outcomes considered for PRAF and RPS being detected, but with more complex and nuanced patterns of response emerging in the case of PROGRESA. The overall results indicate that the programs have not introduced any substantial disincentives to work and that they have had no significant effect on the intensive or the extensive margin of labor supply for individuals or households in treatment localities. The finding that substantial monetary transfers have not had an impact on employment is compatible with a setting in which income effects (assuming that leisure is a normal good for beneficiaries) are either small or are counterbalanced by some of the other factors discussed in Section 2. This seems to be the case, for instance, with regard to the small (about 0.4 hours per week) but significant increase in the number of hours of work for female beneficiaries in PROGRESA. This increase in the intensive margin of the labor supply for working women is compatible with the existence of some of the channels that may link CCTs with an increasing labor supply. More specifically, this effect can be associated with the program's positive effect on the school enrollment rate for children in eligible households, which in turn may have increased the amount of time that women have available to devote to paid employment. Indeed, previous evidence for the program found that it did, in fact, reduce women's participation in domestic work (Parker and Skoufias, 2000).

The results for PROGRESA, however, point to the presence of more complex effects on labor-market outcomes. Angelucci and De Giorgi (2009) document the various types of positive welfare effects that PROGRESA has had on ineligible households in program localities; for instance, they report higher consumption, which seems to be mediated by credit markets, gifts and the overall effect of the program on the local economy. This study confirms the presence of these spillover and equilibrium effects and provides additional evidence in the form of the impact in terms of the labor-market outcomes for adults. The indirect treatment effects signal a reduction in employment, which is mostly driven by ineligible females. This contrasts with the increase in the number of hours worked by eligible females, although the latter effect is too small to suggest that eligible women fully displaced their ineligible counterparts from their jobs. The lack of a substantial effect with regard to the number of hours worked or household labor income per adult for the ineligible population indicates the presence of some reallocation of labor within ineligible households. Moreover, there was a substantially greater shift toward agricultural employment on the part of ineligible males than there was among their eligible counterparts. On the other hand, the program substantially increased the wages and labor income of males in eligible households. This evidence, taken as a whole, suggests that PROGRESA allowed eligible males to move away from agricultural work and toward higher-paying employment, with neutral sector reallocation at the locality level (as indicated by the lack of significant aggregate effects on agricultural employment). The evidence concerning the relative shift of ineligibles toward agricultural labor is consistent with the finding of Skoufias et al. (2008) that the PAL program in Mexico induced workers to move away from agricultural work. It also further supports the idea that employment in agriculture acts as food insurance. Eligible individuals in PROGRESA seemed to be able to take advantage of more risky but potentially more rewarding non-agricultural employment opportunities, or at least to do so more than their ineligible counterparts could.

Taken as a whole, the evidence indicates that, while CCT programs in poor rural areas with high benefit levels do not create a major disincentive to work, they may still have effects in terms of employment-related outcomes and may nonetheless influence the equilibrium of the labor market. The presence of equilibrium effects in this context should not be surprising: for example, in the first stage of PROGRESA's implementation, about half of the households in treatment localities received a transfer equivalent to 40 percent of their income. The aggregate effect of such large, widespread transfers must have had an impact at the community level, above and beyond their effect on eligible households, and this is reflected in Angelucci's and De Giorgi's (2009) finding that PROGRESA has had a positive effect on the consumption levels of ineligible households.

These results have important implications for the evaluation, design, implementation and scope of future programs. Equilibrium effects complicate the interpretation of reduced form estimates from randomized controlled experiments, which is a long-standing discussion in regard to the analysis of welfare programs (see, for instance, Browning's (1971) critique of Orcutt's and Orcutt's (1968) randomized negative income tax experiments based on feedback effects of wages). In terms of the empirical results presented here, the programs' impacts can be attributed to shifts in sector allocation and access to better income-generating opportunities for males in eligible households and to the increase in the amount of time available to women associated with higher school enrollment rates for children. It is not possible, however, to rule out further effects linked to program-induced aggregate changes in labor demand. Moreover, in the case of CCTs whose intended outcomes span multiple dimensions, indirect effects on labor-market outcomes could be confounded with the direct impact of the transfers and the programs' conditionalities. The results suggest that labor-market effects, apart from work disincentives, should be taken into account in the design of welfare programs in developing countries and that the evaluations of such programs should seek to disentangle the underlying mechanisms that are at work.

References

- Alzua M, Cruces, G, Ripani L (2010) Welfare programs and labor supply in developing countries: Evidence from Latin America. CEDLAS Working Paper 95.
- Angelucci M, De Giorgi G (2009) Indirect Effects of an Aid Program: How Do Cash Transfers Affect Ineligibles' Consumption? Am Econ Rev 99:1, pp 486–508.
- Angrist J, Imbens G, Rubin D (1996) Identification of Causal Effects Using Instrumental Variables. J Am Statistical Assoc vol. 91, No. 434, pp. 444-455.
- Angrist J, Krueger A (1999) Empirical Strategies in Labor Economics. In Ashenfelter O, Card D (eds) Handbook of Labor Economics, vol. 3A, 2097 North-Holland.
- Angrist J, Pischke J (2008) Mostly Harmless Econometrics: An Empiricist's Companion, Princeton University Press.
- Ardington C, Case A, Hosegood V (2009) Labor Supply Responses to Large Social Transfers: Longitudinal Evidence from South Africa, Am Econ J: Appl Econ vol. 1, No. 1.
- Ashenfelter O, Plant M (1990) Nonparametric Estimates of the Labor-Supply Effects of Negative Income Tax Programs. J Labor Econ vol. 8, No. 1.
- Baker M, Gruber J, Milligan K (2008) Universal Child Care, Maternal Labor Supply, and Family Well-Being. J Polit Econ vol. 116(4), pp. 709-745.
- Basu K, Hoang Van P (1998) The Economics of Child Labor. Am Econ Rev vol. 88, No. 3.
- Behrman J, Hoddinott J (2005) Programme Evaluation with Unobserved Heterogeneity and Selective Implementation: The Mexican "PROGRESA" Impact on Child Nutrition. Oxf Bull Econ and Stat vol. 67(4).
- Behrman J, Todd P (1999) Randomness in the Experimental Samples of PROGRESA (Education, Health and Nutrition Program). Report submitted to PROGRESA, International Food Policy Research Institute, Washington D.C.
- Bertrand M, Duflo E, Mullainathan S (2004) How Much Should We Trust Differences in Differences Estimates? Q J Econ vol. 119(1), pp. 249-275.
- Besley T, Coate S (1992) Workfare versus Welfare: Incentive Arguments for Work Requirements in Poverty Alleviation Programs. Am Econ Rev vol. 82, No. 1, pp. 249-261.
- Bhattarai K, Whalley J (2003) Discreteness and the Welfare Cost of Labor Supply Tax Distortions. Int Econ Rev vol. 44(3), pp. 1117-1133.
- Blau D, Tekin E (2007) The Determinants and Consequences of Child Care Subsidies for Single Mothers in the USA. J Population Econ vol. 20(4), pp. 719-741.
- Blundell R, Hoynes H (2004) Has 'In-Work' Benefit Reform Helped the Labor Market? In NBER Seeking a Premier Economy: The Economic Effects of British Economic Reforms, 1980-2000, pp. 411-460, National Bureau of Economic Research.
- Blundell R, MaCurdy T (1999) Labor Supply: A Review of Alternative Approaches. In Ashenfelter O, Card D (eds.) Handbook of Labor Economics vol. 3A, 2097 North-Holland.

- Bobonis G, Finan F (2009) Neighborhood Peer Effects in Secondary School Enrollment Decisions. Rev Econ Statist 91(4), pp. 695-716.
- Bouillon C, Tejerina L (2006) Do We Know What Works? A Systematic Review of Impact Evaluations of Social Programs in Latin America and the Caribbean. Inter-American Development Bank (IDB), Sustainable Development Department, Poverty and Inequality Unit.
- Browning E (1971) Incentive and Disincentive Experimentation for Income Maintenance Policy Purposes: Note. Am Econ Rev vol. 61, No. 4.
- Cameron A, Gelbach J, Miller D (2008) Bootstrap-Based Improvements for Inference with Clustered Errors. Rev Econ Statist vol. 90(3), pp. 414-427.
- CEDLAS (2012) A Guide to the SEDLAC: Socioeconomic Database for Latin America and the Caribbean. Technical report, Centro de Estudios Distributivos, Laborales y Socials, Universidad Nacional de La Plata, and the World Bank. <u>http://sedlac.econo.unlp.edu.ar/eng/methodology.php</u>.
- Cogan J (1981) Fixed Costs and Labor Supply. Econometrica vol. 49(4), pp. 945-963.
- Deaton A (2009) Instruments of development: randomization in the tropics, and the search for the elusive keys to economic development. NBER Working Paper 14690.
- Dickens R., Gregg P, Wadsworth J (eds.) (2004) *The Labour Market Under New Labour: the State of Working Britain.* Palgrave Macmillan, London.
- Djebbari H, Smith J (2008) Heterogeneous Impacts in PROGRESA. J Econometrics vol. 145(1-2), pp. 64-80.
- Donald S, Lang K (2007) Inference with Differences in Differences and Other Panel Data. Rev Econ Statist vol. 89(2), pp. 221-233.
- Duflo E, Glennerster R, Kremer M (2008) Using Randomization in Development Economics Research: A Toolkit. Handbook of Development Economics, Elsevier.
- Eissa N, Kleven H, Kreiner C (2008) Evaluation of Four Tax Reforms in the United States: Labor Supply and Welfare Effects for Single Mothers. J Public Econ vol. 92, pp. 3-4.
- Eissa N, Liebman J (1996) Labor Supply Response to the Earned Income Tax Credit. Q J Econ vol. 111(2), pp. 605-637.
- Fiszbein A, Schady N (2009) Conditional Cash Transfers: Reducing Present and Future Poverty. World Bank, Washington, DC.
- Galiani S, McEwan P (2012). The Heterogeneous Impact of Conditional Cash Transfers in Honduras. <u>http://ssrn.com/abstract=1931216</u>.
- Gertler P (2004) Do Conditional Cash Transfers Improve Child Health? Evidence from PROGRESA's Control Randomized Experiment. Am Econ Rev vol. 94(2), pp. 336-341.
- Glewwe P, Olinto P (2004) Evaluating the Impact of Conditional Cash Transfers on Schooling: An Experimental Analysis of Honduras's PRAF Program. Final Report for USAID.
- Heckman J (2008) Econometric Causality. Int Statistical Rev vol. 76(1), pp. 1-27, 04.
- IFPRI (2000) PRAF, Third Report: Monitoring and Evaluation System. IFPRI, Washington DC.

- Imbens G (2010) Better LATE Than Nothing: Some Comments on Deaton (2009) and Heckman and Urzua (2009). J Econ Lit vol 48(2), pp. 399–423.
- Imbens G, Rubin D, Sacerdote B (2001) Estimating the Effect of Unearned Income on Labor Earnings, Savings and Consumption: Evidence from a Survey of Lottery Players. Am Econ Rev vol. 91(4), pp. 779-794.
- Kanbur R, Keen M, Tuomala M (1994) Labor Supply and Targeting in Poverty Alleviation Programs. World Bank Econ Rev vol. 8(2), pp. 191-211.
- Kluve J, Tamm M (2012) Parental leave regulations, mothers' labor force attachment and fathers' childcare involvement: evidence from a natural experiment. J Popul Econ. doi: 10.1007/s00148-012-0404-1
- Maluccio J (2004) Effects of Conditional Cash Transfer Programs on Current Poverty, Consumption, and Nutrition. Paper presented at the Second International Workshop on Conditional Cash Transfer Programs, São Paulo, Brazil, April 2004.
- Maluccio J (2007) The Impact of Conditional Cash Transfers in Nicaragua on Consumption, Productive Investments, and Labor Allocation. ESA Working Paper No. 07-11, Agricultural and Development Economics Division, United Nations Food and Agriculture Organization (FAO).
- Maluccio J, Flores, R (2005) Impact Evaluation of a Conditional Cash Transfer Program: the Nicaraguan Red de Protección Social. Report 141, International Food Policy Research Institute, Washington, DC.
- Meghir C, Phillips D (2008) Labour Supply And Taxes. Working Paper 008/04, Institute For Fiscal Studies, London.
- Michalopoulos C, Robins P, Card D (2005) When Financial Work Incentives Pay for Themselves: Evidence from a Randomized Social Experiment for Welfare Recipients. J Public Econ vol. 89(1), pp. 5-29.
- Moffitt R (2002) Welfare Programs and Labor Supply. In Auerbach A, Feldstein M. (eds.), Handbook of Public Economics, edition 1, vol. 4, Chapter 34, Elsevier.
- Moffitt R (2003a) The Negative Income Tax and the Evolution of U.S. Welfare Policy. J Econ Perspect vol. 17(3), pp. 119-140.
- Moffitt R (2003b) The Role of Randomized Field Trials in Social Science Research: A Perspective from Evaluations of Reforms of Social Welfare Programs. NBER Technical Working Paper 0295, National Bureau of Economic Research.
- Moffitt R, Scholz J (2009) Trends in the Level and Distribution of Income Support. NBER Working Paper 15488, National Bureau of Economic Research.
- Mörk E, Sjögren A, Svaleryd H (2011) Childcare costs and the demand for childrenevidence from a nationwide reform. J Popul Econ. doi: 10.1007/s00148-011-0399-z
- Orcutt G, Orcutt A (1968) Incentive and Disincentive Experimentation for Income Maintenance Policy Purposes. Am Econ Rev vol. 58, No. 4.
- Parker S, Skoufias E (2000) The Impact of PROGRESA on Work, Leisure, and Time Allocation. Final Report, International Food Policy Research Institute. <u>www.ifpri.org/themes/Progresa/timeallocation.htm</u>. Accessed June 2008.
- Ralitza D, Wolff FC (2011) Do downward private transfers enhance maternal labor supply? Evidence from around Europe. J Popul Econ 24: 911-933

- Rawlings L, Rubio G (2003) Evaluating the Impact of Conditional Cash Transfer Programs: Lessons from Latin America. World Bank Policy Research Working Paper No. 3119.
- Rawlings L, Rubio G (2005) Evaluating the Impact of Conditional Cash Transfer Programs. World Bank Res Observer vol. 20(1), pp. 29–55.
- Skoufias E, Davis B, Behrman J (1999) An Evaluation of the Selection of Beneficiary Households in the Education, Health, and Nutrition Program (PROGRESA) of Mexico. Report submitted to PROGRESA, International Food Policy Research Institute, Washington, D.C.
- Skoufias E, Di Maro V (2008) Conditional Cash Transfers, Adult Work Incentives, and Poverty. J Devel Stud vol. 44(7), pp. 935–960.
- Skoufias E, Parker S (2001) Conditional Cash Transfers and Their Impact on Child Work and Schooling: Evidence from the PROGRESA Program in Mexico. Economía vol. 2, No. 1, Fall 2001.
- Skoufias E, Parker S (2006) Job Loss and Family Adjustments in Work and Schooling during the Mexican Peso Crisis. J Population Econ vol. 19(1), pp. 163-181.
- Skoufias E, Unar M, González-Cossío T (2008) The Impacts of Cash and In-Kind Transfers on Consumption and Labor Supply: Experimental Evidence from Rural Mexico. Policy Research Working Paper 4778, World Bank.
- Todd J, Winters P, Stecklov G (2011) Evaluating the Impact of Conditional Cash Transfer Programs on Fertility: The Case of the Red de Protección Social in Nicaragua. J Population Econ vol. 25(1), pp. 267-290.
- Todd P (2004) Design of the Evaluation and Method Used to Select Comparison Group Localities for the Six-Year Follow-up Evaluation of Oportunidades in Rural Areas. Technical Note, Social Development Secretariat (SEDESOL), Mexico.
- Wooldridge J (2001) Econometric Analysis of Cross Section and Panel Data. MIT Press.
- Wooldridge J (2007) Difference-in-Differences Estimation. Lecture Notes 10, Guido Imbens and James Wooldridge course, What's New in Econometrics, NBER, Summer.
- Yang D (2008) International Migration, Remittances and Household Investment: Evidence from Philippine Migrants' Exchange Rate Shocks. Econ J vol. 118(528).

Table 1 **Descriptive Statistics, by Program**

	PRAF (Base Dec. 2	-	PROGRESA Sept. 97-		RPS (Basel Sept. 2	-
	Treatment	Control	Treatment	Control	Treatment	Control
<u>Households</u>						
All	34,455	29,538	14,856	9,221	5,781	4,203
Eligible (%)			52.8	50.8		
Spouse present (%)	77.4	73.9	81.9	82.3	81.4	79.4
	(1.06)	(1.10)	(0.32)	(0.40)	(1.37)	(1.46)
Mean household size	6.6	6.6	6.5	6.5	7.5	7.5
	(0.03)	(0.03)	(0.01)	(0.01)	(0.05)	(0.05)
Mean number of children	3.4	3.2	3.3	3.2	3.3	3.3
	(0.03)	(0.03)	(0.01)	(0.01)	(0.04)	(0.03)
Individuals Years of Education						
All	3.4	2.9	5.2	5.1	2.3	2.2
	(0.05)	(0.04)	(0.02)	(0.02)	(0.05)	(0.05)
Females	3.4	2.7	5.1	5.1	2.3	2.3
	(0.07)	(0.06)	(0.02)	(0.03)	(0.07)	(0.07)
Males	3.4	3.1	5.2	5.1	2.2	2.1
	(0.06)	(0.06)	(0.02)	(0.03)	(0.07)	(0.07)
Employment						
All	66.3	64.5	53.1	51.2	56.2	57.7
	(0.78)	(0.76)	(0.24)	(0.30)	(0.98)	(1.00)
Females	34.4	32.3	19.5	16.8	20.4	23.2
	(1.22)	(1.13)	(0.27)	(0.31)	(1.15)	(1.22)
Males	90.0	88.3	86.4	85.7	90.5	91.7
	(0.64)	(0.67)	(0.23)	(0.30)	(0.81)	(0.79)
Households with spouse	66.0	65.3	52.4	50.8	56.0	57.1
	(0.87)	(0.85)	(0.26)	(0.32)	(1.07)	(1.10)
Households without spouse	67.4	61.3	57.4	53.5	57.0	60.5
	(1.77)	(1.64)	(0.62)	(0.79)	(2.54)	(2.42)
Households with children	67.2	66.1	53.2	51.6	56.8	58.4
	(0.87)	(0.86)	(0.27)	(0.34)	(1.11)	(1.13)
Households without children	62.9	59.3	52.8	49.9	53.8	55.3
	(1.75)	(1.59)	(0.49)	(0.61)	(2.13)	(2.19)
Agricultural Workers						
All	65.3	64.4	76.9	74.7	83.7	80.9
	(0.98)	(0.95)	(0.35)	(0.44)	(0.98)	(1.06)
Females	33.0	33.8	41.8	32.7	57.2	42.2
	(2.32)	(2.02)	(1.09)	(1.28)	(3.25)	(3.11)
Males	74.5	72.7	82.6	81.3	89.2	90.5
	(0.99)	(0.99)	(0.34)	(0.42)	(0.90)	(0.88)
Mean hours of work						
All	38.2	37.7	43.4	43.7	39.4	38.0
	(0.21)	(0.20)	(0.10)	(0.12)	(0.42)	(0.44)
Females	33.0	32.4	41.5	42.3	33.1	31.2
	(0.69)	(0.59)	(0.28)	(0.39)	(1.28)	(1.17)
Males	39.6	39.1	43.8	44.0	41.0	40.2
	(0.19)	(0.19)	(0.10)	(0.12)	(0.40)	(0.43)

Source: Own calculations based on program evaluation surveys. Standard errors are shown in parentheses.

Table 2 **Program Effect on Employment: PRAF and RPS**

			DD Est	timates		
	П	Т	ITT N	Nales	ITT F€	emales
PRAF	OLS	FE	OLS	FE	OLS	FE
Baseline: Aug-Dec. 2000						
t=1 (May-Aug. 2002)	-0.011	-0.015	-0.005	-0.012	-0.010	-0.018
	(0.016)	(0.015)	(0.018)	(0.018)	(0.023)	(0.028)
Observations	12,833	12,482	7,145	6,930	5,688	5,552
Groups		7,484		3,918		3,569
RPS Baseline: Aug-Sept. 2000						
t=1 (Oct. 2001)	-0.005	-0.002	0.009	0.006	-0.022	-0.010
	(0.020)	(0.020)	(0.020)	(0.020)	(0.030)	(0.029)
t=2 (Oct. 2002)	-0.012	-0.013	-0.009	-0.005	-0.020	-0.023
	(0.018)	(0.019)	(0.014)	(0.014)	(0.030)	(0.031)
Observations	11,241	11,287	5,828	5,852	5,413	5,435
Groups		4,426		2,300		2,126

Source: Own calculations based on program evaluation surveys. Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

Table 3 **Program Effect on Employment: PROGRESA**

									DD Est	timates								
PROGRESA	П	Т	A	ΓE	IT	E		/lales	ITT Fe	emales	ATE	Males	ATE Fe	emales	ITEN	Males	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE
t=1 (Nov. 98)	-0.008	-0.008	-0.003	-0.003	-0.015	-0.015	-0.002	0.003	-0.012	-0.010	0.008	0.005	-0.008	-0.001	-0.012	-0.009	-0.020	-0.019
	(0.008)	(0.008)	(0.010)	(0.010)	(0.009)	(0.009)*	(0.009)	(0.008)	(0.015)	(0.013)	(0.012)	(0.010)	(0.019)	(0.016)	(0.011)	(0.011)	(0.017)	(0.017)
t=2 (Mar. 99)	-0.014	-0.009	-0.012	-0.009	-0.026	-0.014	0.000	0.004	-0.024	-0.020	0.002	0.000	-0.019	-0.021	-0.002	-0.001	-0.046	-0.020
	(0.007)**	(0.007)	(0.009)	(0.009)	(0.011)**	(0.010)	(0.009)	(0.007)	(0.016)	(0.011)*	(0.011)	(0.009)	(0.017)	(0.015)	(0.014)	(0.012)	(0.021)**	* (0.017)
t=3 (Nov. 99)	-0.009	-0.006	-0.005	-0.005	-0.023	-0.016	0.003	-0.001	-0.020	-0.010	0.005	-0.002	-0.011	-0.003	-0.004	-0.009	-0.049	-0.035
	(0.007)	(0.007)	(0.009)	(0.009)	(0.011)**	(0.010)*	(0.009)	(0.006)	(0.015)	(0.013)	(0.011)	(0.008)	(0.017)	(0.016)	(0.012)	(0.012)	(0.020)**	* (0.020)*
Observations	227,619	232,725	141,271	143,347	84,052	87,044	114,462	114,837	113,157	113,587	70,812	71,013	70,459	70,697	42,545	42,700	41,507	41,680
Groups		72,933		50,636		34,344		61,144		60,632		40,646		40,304		26,401		26,090

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

			DD Est	timates		
	П	Т		<i>N</i> ales	ITT Fe	emales
PRAF	OLS	FE	OLS	FE	OLS	FE
Baseline: Aug-Dec. 2000						
=1 (May-Aug. 2002)	-0.028	-0.030	-0.030	-0.040	-0.036	0.010
	(0.039)	(0.039)	(0.041)	(0.043)	(0.057)	(0.047)
Observations	8,158	7,931	6,451	6,257	1,707	1,674
Groups		5,034		3,746	`	1,289
RPS						
Baseline: Aug-Sept. 2000						
=1 (Oct. 2001)	-0.002	0.017	0.001	0.014	-0.004	0.041
	(0.024)	(0.020)	(0.023)	(0.020)	(0.078)	(0.078)
=2 (Oct. 2002)	-0.013	0.016	-0.002	0.010	-0.037	0.083
. ,	(0.022)	(0.019)	(0.021)	(0.019)	(0.068)	(0.063)
Observations	6,438	6,464	5,484	5,505	954	959
Groups		2,903		2,239		664

Table 4 Program Effect on Agricultural Employment: PRAF and RPS

Source: Own calculations based on program evaluation surveys. Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

									DD Est	timates								
PROGRESA	П	Т	A	TE	П	E		<i>l</i> ales	ITT Fe	emales	ATE	Males	ATE F	emales	ITEI	Vales	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE
t=1 (Nov. 98)	-0.010	-0.007	-0.015	-0.010	-0.005	0.000	-0.010	-0.006	-0.033	-0.030	-0.014	-0.009	-0.061	-0.057	-0.008	-0.001	-0.009	-0.022
	(0.012)	(0.010)	(0.011)	(0.011)	(0.016)	(0.016)	(0.012)	(0.012)	(0.035)	(0.032)	(0.013)	(0.013)	(0.047)	(0.055)	(0.018)	(0.020)	(0.038)	(0.044)
t=2 (Mar. 99)	-0.007	-0.001	-0.009	-0.004	0.011	0.029	-0.004	0.002	-0.031	-0.025	-0.007	-0.008	-0.056	-0.053	0.019	0.061	0.002	-0.015
	(0.011)	(0.010)	(0.012)	(0.011)	(0.020)	(0.020)	(0.012)	(0.012)	(0.034)	(0.033)	(0.013)	(0.014)	(0.046)	(0.054)	(0.025)	(0.025)**	(0.040)	(0.050)
t=3 (Nov. 99)	0.004	0.003	0.007	0.001	0.004	0.019	0.016	0.010	-0.031	-0.027	0.015	0.004	-0.046	-0.049	0.020	0.045	-0.019	-0.020
	(0.012)	(0.011)	(0.012)	(0.012)	(0.020)	(0.021)	(0.013)	(0.014)	(0.035)	(0.033)	(0.013)	(0.015)	(0.047)	(0.054)	(0.023)	(0.026)*	(0.040)	(0.050)
Observations	84,210	86,176	52,978	53,775	30,208	31,364	51,065	51,210	33,145	33,269	31,576	31,641	21,402	21,463	18,917	18,990	11,291	11,348
Groups		38,583		25,200		17,534		30,336		24,646		19,280		16,051		13,276		9,309

Table 5 Program Effect on Agricultural Employment: PROGRESA

Source: Own calculations based on program evaluation surveys. Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

			DD Est	timates		
	П	Т	ITT N	<i>l</i> ales	ITT Fe	emales
PRAF	OLS	FE	OLS	FE	OLS	FE
Baseline: Aug-Dec. 2000						
t=1 (May-Aug. 2002)	0.681	0.814	0.493	0.580	1.840	1.849
	(0.644)	(0.650)	(0.621)	(0.617)	(1.335)	(1.716)
Observations	8,139	7,913	6,438	6,245	1,701	1,668
Groups		5,029		3,745		1,285
RPS						
Baseline: Aug-Sept. 2000						
t=1 (Oct. 2001)	-2.638	-2.982	-2.261	-2.667	-3.030	-4.067
	(1.846)	(1.807)	(1.620)	(1.649)	(4.473)	(4.734)
t=2 (Oct. 2002)	-1.996	-1.971	-1.475	-1.672	-5.668	-4.001
. ,	(1.890)	(1.882)	(1.799)	(1.798)	(4.116)	(4.584)
Observations	6,634	6,660	5,503	5,524	1,131	1,136
Groups		3,021		2,245		776

Table 6 Program Effect on Individual Hours of Work per Week: PRAF and RPS

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

									DD Es	stimates								
PROGRESA	П	Т	A	ΓE	IT	E	ITT N	<i>l</i> ales	ITT Fe	emales	ATE	Males	ATE F	emales	ITE N	Males	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE
t=1 (Nov. 98)	0.019 (0.051)	-0.005 (0.047)	0.051 (0.058)	0.005 (0.056)	-0.022 (0.065)	-0.022 (0.058)	0.000 (0.053)	0.053 (0.058)	0.136 (0.109)	0.256 (0.167)	-0.014 (0.067)	0.026 (0.069)	0.417 (0.157)***	0.484 * (0.276)*	0.024 (0.068)	0.078 (0.076)		0.152 (0.236)
t=3 (Nov. 99)	0.037 (0.053)	0.055 (0.051)	0.055 (0.062)	0.026 (0.063)	-0.012 (0.075)	0.036 (0.079)	-0.015 (0.056)	0.037 (0.057)	0.184 (0.110)*	0.353 (0.172)**	-0.005 (0.067)	0.004 (0.069)	0.409 (0.157)***	0.474 * (0.273)*	-0.025 (0.084)	-0.037 (0.106)	0.001 (0.143)	0.384 (0.246)
Observations	78,236	80,078	44,077	44,816	33,268	34,361	50,503	50,609	27,733	27,831	27,923	27,963	16,154	16,201	22,128	22,188	11,140	11,187
Groups		41,813		24,951		20,662		33,167		23,652		18,832		13,957		16,180		10,026

Table 7 Program Effect on Individual Hours of Work per Week: PROGRESA

Source: Own calculations based on program evaluation surveys. Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

Program Effect on Hours	worked by			, 1	I: PKAF and	u KPS
				timates		
	П	Т		Males	ITT Fe	emales
PRAF	OLS	FE	OLS	FE	OLS	FE
Baseline: Aug-Dec. 2000						
t=1 (May-Aug. 2002)	0.451	0.304	0.466	0.331	0.871	0.122
	(0.572)	(0.609)	(0.604)	(0.648)	(1.166)	(1.326)
Observations	5,344	5,344	4,537	4,537	807	807
Groups		2,999		2,540		525
RPS						
Baseline: Aug-Sept. 2000						
t=1 (Oct. 2001)	-1.872	-1.938	-1.835	-1.903	-1.690	-2.103
	(1.157)	(1.157)	(1.128)	(1.129)*	(3.336)	(3.313)
t=2 (Oct. 2002)	-1.602	-1.559	-1.559	-1.460	-1.869	-2.541
-	(1.096)	(1.099)	(1.092)	(1.080)	(2.647)	(2.894)
Observations	4,124	4,124	3,652	3,652	472	472
Groups		1,525		1,331		194

Table 8 Program Effect on Hours Worked by Adults in the Household ner Adult: PRAF and RPS

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

Table 9Program Effect on Hours Worked by Adults in the Household, per Adult: PROGRESA

									DD Est	timates								
PROGRESA	П	Т	A	ΓE	IT	E	ITT N	/lales	ITT Fe	emales	ATE	Males	ATE F	emales	ITE I	Males	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE																
t=1 (Nov. 98)	-0.094	0.098	-0.262	-0.113	-0.011	0.078	-0.199	0.159	0.117	0.974	-0.495	-0.334	0.007	-1.441	0.054	0.481	0.277	1.890
	(0.450)	(0.446)	(0.547)	(0.547)	(0.556)	(0.539)	(0.486)	(0.501)	(0.825)	(1.241)	(0.581)	(0.604)	(1.237)	(1.831)	(0.639)	(0.657)	(1.095)	(1.781)
t=3 (Nov. 99)	-0.384	-0.149	-0.374	-0.436	-0.886	-0.497	-0.357	-0.143	-0.288	0.723	-0.294	-0.486	-0.371	-2.222	-0.873	-0.215	-0.415	1.938
	(0.483)	(0.492)	(0.575)	(0.621)	(0.650)	(0.689)	(0.504)	(0.522)	(0.858)	(1.220)	(0.586)	(0.638)	(1.256)	(1.918)	(0.758)	(0.902)	(1.244)	(1.796)
Observations	55,973	57,180	34,255	34,775	20,893	21,580	37,483	37,483	18,490	18,490	22,348	22,348	11,907	11,907	14,719	14,719	6,174	6,174
Groups		25,148		17,024		11,147		22,871		14,661		14,242		9,745		10,093		5,241

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses.

* Significant at 10%; ** significant at 5%; *** significant at 1%.

Table 10 **Program Effect on Individual Log Hourly Wages: PROGRESA**

									DD E	stimates								
PROGRESA	17	ΓT	A	TE	IT	E	ITT N	Males	ITT Fe	emales	ATE	Males	ATE F	emales	ITE N	Aales	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE
t=1 (Nov. 98)	0,011	0,015	-0,005	0,007	0,030	0,024	0,012	0,015	-0,022	-0,094	0,003	0,011	-0,100	-0,097	0,027	0,034	0,042	-0,049
	(0,032)	(0,030)	(0,040)	(0,034)	(0,037)	(0,039)	(0,031)	(0,033)	(0,090)	(0,074)	(0,037)	(0,041)	(0,164)	(0,130)	(0,039)	(0,042)	(0,075)	(0,101)
t=3 (Nov. 99)	0,051	0,057	0,050	0,069	0,052	0,036	0,054	0,075	0,011	-0,089	0,056	0,098	-0,045	-0,069	0,061	0,045	0,033	-0,068
	(0,034)	(0.030)*	(0,040)	(0.035)**	(0,040)	(0,040)	(0.031)*	(0.035)**	(0,092)	(0,074)	(0,035)	(0.042)**	(0,164)	(0,137)	(0,042)	(0,052)	(0,077)	(0,100)
Observations	71.536	73.316	41.259	41.977	29.400	30.453	46.150	46.252	25.386	25.478	26.131	26.171	15.128	15.174	19.575	19.631	9.825	9.868
Groups		38.917		23.740		18.704		30.796		21.587		17.900		13.046		14.576		8.830

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses.

* Significant at 10%; ** significant at 5%; *** significant at 1%.

Table 11 Program Effect on Log Household Labor Income, per Adult: PROGRESA

									DD Est	timates								
PROGRESA	רו	Т	A	TE	IT	Ē	ITT I	Males	ITT Fe	emales	ATE	Males	ATE F	emales	ITE N	Males	ITE Fe	emales
Baseline: Sept. 97-Mar. 98	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE	OLS	FE
t=1 (Nov. 98)	0,003	0,016	-0,013	0,004	0,024	0,036	-0,007	0,002	0,038	0,093	-0,019	-0,002	-0,027	0,054	0,010	0,005	0,114	0,075
	(0,021)	(0,020)	(0,021)	(0,021)	(0,031)	(0,030)	(0,023)	(0,024)	(0,051)	(0,065)	(0,023)	(0,025)	(0,063)	(0,090)	(0,035)	(0,039)	(0,075)	(0,099)
t=2 (Mar. 99)	0,020	0,027	0,012	0,023	0,036	0,036	0,024	0,021	0,037	0,093	0,022	0,031	-0,020	0,064	0,030	-0,004	0,108	0,053
	(0,023)	(0,022)	(0,023)	(0,023)	(0,038)	(0,039)	(0,024)	(0,025)	(0,052)	(0,066)	(0,026)	(0,028)	(0,065)	(0,092)	(0,044)	(0,048)	(0,079)	(0,109)
t=3 (Nov. 99)	0,027	0,039	0,031	0,046	-0,001	-0,006	0,018	0,031	0,061	0,097	0,026	0,052	0,014	0,083	-0,023	-0,045	0,092	0,008
	(0,024)	(0.023)*	(0,024)	(0.024)*	(0,037)	(0,038)	(0,024)	(0,025)	(0,051)	(0,064)	(0,025)	(0.027)*	(0,065)	(0,091)	(0,043)	(0,049)	(0,076)	(0,105)
Observations	65.500	66.654	43.181	43.701	21.431	22.065	41.623	41.623	23.877	23.877	26.569	26.569	16.612	16.612	14.563	14.563	6.868	6.868
Groups		24.781		17.451		10.701		22.350		16.490		14.811		11.706		9.513		5.366

Source: Own calculations based on program evaluation surveys.

Standard errors, clustered at the locality level, are shown in parentheses. * Significant at 10%; ** significant at 5%; *** significant at 1%.

Online appendix

Appendix: A Brief Description of the Programs

Mexico: PROGRESA Program

In 1997, Mexico began implementing the first phase of the PROGRESA (later renamed the *Oportunidades*) conditional cash transfer program in rural areas. The program had a multisector focus, with intended impacts on education, health and nutrition, in addition to the potential for poverty alleviation as a result of the cash transfer itself.

The initial deployment of the program was designed to facilitate the evaluation of its impact. The program was geographically targeted by locality, based on a poverty index. From an initial group of 506 localities that were selected for the first round, 320 were randomly selected to participate in the PROGRESA program (i.e., qualifying households in those localities would be eligible to participate); the program was not deployed in the remaining 186 localities. Households in the latter localities were still subject to the data collection process and thus constituted the control group for the program's evaluation.¹ Although the program was later expanded to cover additional areas, this study has focused on the initial deployment stage (1997-1999).

The data employed in this study have been drawn from the PROGRESA Evaluation Survey (ENCEL-*Encuesta de Evaluación de los Hogares*). The estimates given here are based on the initial baseline survey and three follow-up rounds that were conducted on a biannual basis following the program's introduction. The surveys collected socio-demographic and labor-market information at the household and individual levels for both the treatment and control localities.

Honduras: PRAF Program

The *Programa de Asignación Familiar* ("Family Allowance Program") (PRAF) was created by the Government of Honduras in the early 1990s as a compensatory mechanism to mitigate the impact of macroeconomic adjustments on the poor and to alleviate structural poverty. Its coverage was expanded a number of times until it ultimately reached a target population of 173,000 households with children from 0 to 14 years of age in 2008, and it now constitutes one of the largest

¹ The evaluation followed a phased-in process: PROGRESA was deployed in the control localities when the program's coverage was expanded in 2000.

welfare programs in the country. The objective of the program is to encourage poor households to invest in human capital –primarily education and health– through conditional cash transfers.

This study concentrates on the second phase of the program (PRAF II), which entailed a reorganization of the original intervention that had been planned in the late 1990s (Glewwe and Olinto, 2004). The PRAF II phase was implemented in 2000 and was geographically targeted at the municipality level in the poorest region of the country (IFPRI, 2000). It was deployed in a set of 50 randomly selected municipalities out of a total of 70, with the other 20 municipalities forming the control group. While the experimental design of PROGRESA represented an attempt to facilitate an evaluation of the overall impact of the program, the PRAF design was more ambitious. The original intention was to permit an evaluation of different types of interventions, and three sub-groups were therefore created within the treatment group: (i) municipalities scheduled to receive a demand-side intervention (cash subsidies), (ii) those scheduled to receive a supply-side intervention (i.e., construction of schools and health centers), and (iii) a group that would receive both. The empirical results presented below are based on the control group and the municipalities in the first treatment sub-group, since the supply-side interventions were never implemented (Glewwe and Olinto, 2004), and there are very few municipalities in the combinedintervention group.² A total of 40 municipalities were used for the estimation, with 20 of those municipalities being PRAF-eligible households and the other 20 being ones in which the program was not deployed.

The data used in this study were drawn from a baseline survey carried out in the second half of 2000 and a follow-up survey conducted in 2002, with a reasonably low sample attrition rate of approximately 8 percent. In contrast to the case of PROGRESA, where all households in treatment and control localities were interviewed, the PRAF surveys covered a sample of households. The corresponding sampling weights are used when possible in the empirical work outlined below.

² For more detailed information, see Glewwe and Olinto (2004).

Nicaragua: RPS Program

The *Red de Protección Social* ("Social Protection Network") (RPS) conditional cash transfer program was launched in 2000. The first phase consisted of a three-year pilot in two rural areas of the central region of Nicaragua (Madriz and Matagalpa), where poverty rates are above the national average. The program was broadly modeled after PROGRESA, and its main objective was to improve households' human capital through conditional cash transfers.

The 42 localities ("*comarcas*") with the lowest ratings on a multidimensional marginality index within the intervention area were selected for the pilot. Half of those localities were randomly assigned to the treatment group and the other half to the control group (Maluccio and Flores, 2005). The program was originally scheduled to be deployed in the control group localities after a year, but, due to a series of delays, they were not brought into the program until two years later.

The data used in this document were drawn from the initial baseline survey, which was carried out in the third quarter of 2000, and the first and second followups, which were conducted in October 2001 and October 2002, respectively. The sample attrition rate was approximately 7 percent. As with the PRAF evaluation data, the survey covered a sample of the targeted population, and sampling weights have been used when possible.

Appendix: Analysis of the Random Assignment Process

Tables A1-A3 present the results of a probit regression of the probability of being selected into the treatment sample for each program as a function of observable household and individual characteristics. Since the focus of this study is on estimating differentiated treatment effects across population subgroups, estimates are presented for the entire adult population aged 15-65, as well as separately by gender, with standard errors being clustered at the locality level.³ The results reveal some significant differences between treatment and control groups under all three programs.

The results for PRAF indicate that the situation in the treatment and control localities differed in two important ways. Households in the treatment sample had a significantly higher proportion of children attending school and a significantly lower proportion of children who were employed. There do not seem to be any significant differences by gender.

In localities selected for PROGRESA deployment, there was a significantly higher proportion of individuals who were employed, and this effect was accounted for mainly by men. The treatment and control samples for RPS, according to the probit regression, seem to be more balanced. None of the variables that were included in the analysis appears to be significantly associated with the probability of participation in the program.

³ A number of other models were estimated using other disaggregated characteristics. However, the results were qualitatively similar. Some of the differences disappear once the probit regression accounts for clustering at the locality level.

Variable	All	Men	Women
Age	0.0006	0.0011	0.0005
	(0.001)	(0.001)	(0.001)
Gender (1= Male, 0= Female)	0.0302		
	(0.028)		
Number of children	0.0021	0.0001	0.0028
	(0.009)	(0.009)	(0.009)
Children employed	-0.0865	-0.0930	-0.0813
	(0.0270)***	(0.0297)***	(0.0284)***
Children attending school	0.1032	0.1249	0.0909
	(0.0375)***	(0.0371)***	(0.0400)**
Literacy (1= Yes, 0= No)	0.0019	0.0372	-0.0216
	(0.036)	(0.034)	(0.045)
Employed (1= Yes, 0= No)	-0.0109	-0.0450	0.0427
	(0.035)	(0.046)	(0.038)
Observations	6,897	2,868	4,029
LR Chi2	30.62	30.65	24.16

Table A1-Probit Estimates for Treatment (Marginal Effects), Baseline Year: PRAF

Source: Own calculations based on program evaluation survey.

Standard errors, clustered at the locality level, are shown in parentheses.

* Significant at 10%; ** significant at 5%; *** significant at 1%.

Table A2-Probit Estimates for Treatment (Marginal Effects), Baseline Year: PROGRESA

Variable	All	Men	Women	
Age	-0.0001	0.0002	-0.0003	
-	(0.000)	(0.001)	(0.000)	
Gender (1= Male, 0= Female)	-0.0155			
	(0.011)			
Number of children	0.0038	0.0050	0.0032	
	(0.004)	(0.004)	(0.004)	
Children employed	-0.0113	-0.0192	-0.0067	
	(0.012)	(0.013)	(0.012)	
Children attending school	0.0019	0.0039	-0.0001	
-	(0.015)	(0.016)	(0.015)	
Literacy (1= Yes, 0= No)	-0.0178	-0.0089	-0.0266	
,	(0.026)	(0.029)	(0.025)	
Employed (1= Yes, 0= No)	0.0311	0.0443	0.0160	
	(0.0152)**	(0.0216)**	(0.016)	
Observations	66,646	33,257	33,389	
LR Chi2	9.63	8.71	2.48	

Source: Own calculations based on program evaluation survey.

Standard errors, clustered at the locality level, are shown in parentheses.* Significant at 10%; ** significant at 5%; *** significant at 1%.

Table A3-Probit Estimates for Treatment (Marginal Effects), Baseline Year: RPS

Variable	All	Men	Women	
Age	-0.0001	-0.0001	-0.0001	
	(0.001)	(0.001)	(0.001)	
Gender (1= Male, 0= Female)	0.0217			
	(0.028)			
Number of children	-0.0010	0.0008	-0.0029	
	(0.012)	(0.012)	(0.012)	
Children employed	-0.0509	-0.0623	-0.0390	
	(0.038)	(0.041)	(0.038)	
Children attending school	0.0349	0.0322	0.0388	
	(0.044)	(0.044)	(0.046)	
Literacy (1= Yes, 0= No)	0.0152	0.0118	0.0191	
	(0.033)	(0.043)	(0.032)	
Employed (1= Yes, 0= No)	-0.0188	-0.0292	0.0068	
	(0.035)	(0.046)	(0.049)	
Observations	4,674	2,290	2,384	
LR Chi2	3.96	3.8	2.31	

Source: Own calculations based on program evaluation survey.

Standard errors, clustered at the locality level, are shown in parentheses.* Significant at 10%; ** significant at 5%; *** significant at 1%.

Appendix: Analysis of Hours of Work and Household Composition: RPS

Table A4

Program Effects on Hours Worked by Adults in the Household and on the Number of Adults in the Household: RPS

	Dependent variable: Number of adults in the household							
Baseline: Aug-Sept. 2000	ПТ		ITT Males		ITT Females			
	OLS	FE	OLS	FE	OLS	FE		
t=1 (Oct. 2001)	-0.051	-0.008	-0.027	0.012	-0.295	-0.145		
	(0.045)	(0.033)	(0.044)	(0.032)	(0.137)**	(0.099)		
t=2 (Oct. 2002)	-0.063	-0.057	-0.050	-0.048	-0.190	-0.065		
. ,	(0.055)	(0.039)	(0.056)	(0.039)	(0.178)	(0.172)		
Observations	4,124	4,124	3,652	3,652	472	472		
Groups		1,525		1,331		194		
Baseline: Aug-Sept. 2000	Dependent	variable: Tota	l number of h	ours worked b	by adults in the	e household		
t=1 (Oct. 2001)	-4.444	-5.174	-3.447	-4.356	-12.338	-10.627		
, , ,	(2.925)	(2.787)*	(2.889)	(2.861)	(8.137)	(8.151)		
t=2 (Oct. 2002)	-4.769	-6.380	-3.546	-5.211	-16.023	-14.832		
	(2.823)*	(2.904)**	(2.847)	(2.899)*	(6.565)**	(6.896)**		
Observations	4,124	4,124	3,652	3,652	472	472		
Groups		1,525		1,331		194		

Source: Own calculations based on program evaluation survey.

Standard errors, clustered at the locality level, are shown in parentheses.* Significant at 10%; ** significant at 5%; *** significant at 1%.